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	CRUISE: EX10-04 Leg II
Bitung to Bitung
	PERSONNEL ON BOARD:

NOAA OER:

1. Mashkoor Malik (ERT Inc)

2. Dave Lovalvo (Eastern Oceanics)

3. Webb Pinner (2020 LLC)

4. Jeremy Potter (OER)

Others:

5. Doug Jongeward (PMEL)

6. John Sherrin (UVic)

7. Xerandy (BPPT)

8. Major Muddan (TNI)

9. Dave Wright (UCAR/JOSS)

10. Joel DeMello (UCAR/JOSS)

11. Tom Kok (UCAR/JOSS)

12. Tom Pierce (UCAR/JOSS)

13. Karl McLetchie (UCAR/JOSS)

14. Brian Bingham (UCAR/JOSS)
15. Randy Prickett (UCAR/JOSS)

16. Eric Prechtl (UCAR/JOSS)

17. Roland Brian (UCAR/JOSS)

18. Joe Biscotti (UCAR/JOSS)

	DATE/TIME FILED:  07/9/2010 1122 UTC

FILED BY: Jeremy Potter
	

	VESSEL:  NOAA Ship Okeanos Explorer (EX)


	

	GEOGRAPHIC AREA:  Operating north of North Sulawesi, Indonesia 
	


SUMMARY:    Since the previous SITREP on July 4th, we conducted ROV dives on July 5, July 6, and July 8.  Two CTDs were conducted on the evening of July 6. We completed an additional 6000m cast on July 7. Multibeam operations took place at all other times. 
The video team adapted imagery processing and are implementing workaround to network difficulties to enable ship-to-shore transfer of ROV video and still imagery.
Shore-side and mission personnel completed a draft operations plan that provides for daily ROV dives through July 13.

NOTE: As of today, local ship time is 8 hrs ahead of UTC.  Jakarta is 7 hrs ahead of UTC. East Coast is 4 hrs behind UTC. West Coast is 7 hrs behind UTC.

============================================================================
ROV:
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July 5 -We completed our first ROV dive in the northeast portion of the INDEX-SATAL 2010 area of operations. Our launch location was 5º 22.638'N 126º 46.016’E. During the descent a fuse blew on a lateral thruster. Movement during the dive was hampered but it by no means compromised operations. The ROV reached the seafloor at 2725m and proceeded to climb the slope of a ridge-like feature that peaked at approximately 2600m. The seafloor was covered mostly with fine-grained pelagic sediment. The bottom ranged between mostly sediment with sparse rocky outcrops to steep slopes of broken pillows and talus. Epifauna were generally sparse but we imaged a number of hexactinellid sponges, sea whips, sea stars, some fish, and a variety of corals and sea lillies. We also encountered what appeared to be a very long sea cucumber that was making some interesting tracks in the sediment in search of food. In comparison to previous dives, we noticed fewer crabs and shrimp, but many more sponges. The top of the summit ridge did not yield much different from what we saw during our ridge ascent. The crew replaced the fuse following the dive. No one is quite sure why it blew. Lovalvo believes it was likely due to some of the shock experience during launch in rough seas.
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July 6 - The crew launched the ROV for dive 7 at 5º 14.964'N 126º 39.378’E around 0815. The dive began 150m below the summit on the eastern slope of a steep ridge. The top of the ridge was at a depth of approximately 1050m. The pilots traversed the ROV up a steep face of rock that was lightly covered in fine-grained pelagic sediment and contained several different species of corals, sponges and crinoids. The Team captured excellent close-ups of what may turn out to be a number of completely new species for scientists. Biomass appeared to increase as we approached the summit ridge and was mostly dominated by sessile species. We also happened to spot an octopus hiding beneath a pile of rocks. As the sub traveled in a southerly direction down the ridge, we found an area with relatively few organisms that appeared to have been recently disturbed. We did not find any evidence of hydrothermal or volcanic activity at this site. There were no issues with the lateral thruster. However, after recovering the ROV, a small amount of moisture was discovered in the HD camera housing in the camera platform.  
July 7 - The ROV crew spent the morning examining the moisture issue with the camera sled HD camera. Although the camera system was working fine, it was decided that to expedite the process of getting the ROV back in the water, it was easier install the spare camera. That required that we swap the internal laser between the cameras making the spare compatible with the camera platform.  It is worth noting that any camera issues to date are with Phoenix-provided cameras. The ROV HD camera which is currently in use was purchased by OER and not provide through the Phoenix contract. Since it is unclear to what depth the spare had been pressure tested, the team selected a relatively shallow site for the July 8 dive.
July 8

The crew launched the ROV for dive 8 at 5º 04 678'N 126 º 39.125'E around 0815. The target was a shallow seamount that rises to a depth of approximately 1550m. It lies along a shallow ridge at the southeast corner of the EX1 box. Between landing on bottom and ridge ascent, the topography varied between steep cliffs and relatively level terrain. It was dominated by older, andesitic rocks with a light dusting of pelagic sediment. Corals were abundant in the deeper part of our track but were less so as we approached the summit. Several types of sponges and crinoids became more prominent. Unlike previous dives where sponges were mostly hexactinellids, we saw several different types of sponges with interesting and unique morphologies. Some larger sponges had large, labyrinthian indentations that appeared to be oriented to align with the current. Another type of sponge looked like a furry coconut mat or animal hide.  We did not identify it as a sponge until we harnessed every bit of our powerful high-definition cameras. As we got closer to the summit we came across another octopus that was purple with large glassy eyes. We did not find any hydrothermal vents on this dive but we did succeed in discovering what may turn out to be a number of new species. This dive concluded our exploration of the EX1 box for this leg and we now start our gradual return to Bitung with continued mapping and ROV operations along with way. 
SURVEY/CTD:
EM 302 multibeam mapping continued in the NE priority area. This bathymetry is critical to ROV and CTD target selection. The Expendable Bathymetry Thermograph (XBT) used to provide sound speed profile multibeam sonar suffered a malfunction on July 6. It appears that the XBT connector on fan tail has short circuited due to moisture. A spare part has been ordered and will be hand carried by Meme Lobecker to Indonesia. It will be available for next leg. The Mapping Team continues to use CTD data collected by the ROV to provide sound speed profile for multibeam.
Two CTD casts were conduct at 5º 16.05'N 126 º 37.05'E and 5º 19.61'N 126 º 36.79'E on the evening of July 6. Crew also completed a 6000m cast in the far eastern areas of the Northeast priority box to get a better understanding of regional hydrography. The third CTD was deployed at 5º 52.34'N 127 º 15.20'E
SHIP NETWORK

OMAO shore-side support personnel continue to consider low-speed network issues. No recent progress has been made 
High-speed Network –Since June 20, OER has been paying $6K per day to increase bandwidth from 10Mbps to 20 Mbps to increase the size and quality of the video feeds to the ECCs.  Up until this point the video team and Ship’s ETs have had to reconfigure the network to allow video to be broadcasted at a higher quality.  The ETs have confirmed with MTN that the link to the Okeanos Explorer is 20Mbps and the OMAO network engineer has confirmed the system should support outbound traffic up to 20 Mbps.  Though OMAO is responsible for solving the problem, OER is responsible for paying all MTN contract costs. By July 6th, OER had spent $96K for services that it is unable to use because of the unknown problem.  On that day, the problem apparently disappeared. The team does not consider the problem resolved and continues to work with MTN and NOAA IT to find the root of the problem. 
The ship has started a dialog with NOAANet on connectivity improvements over the high speed link.  These improvements are to enable addition services relate to conducting remote exploration such as remote server management and real-time text-based collaboration. The next step in the process is to gain change approval from OAR CIO and OMAO fleet network security.  OMAO resumed a dialog with the ship and OER personnel to implement changes to network traffic policies.  Progress remains slow, encumbered by personnel availability due to time-zone differences.  There has been no recent progress toward contracting network engineer to examine and trouble-shoot the network from shore. 

Dynamic Positioning

DP has been functioning very well. There are no issues to report.
============================================================================================================================================================
PLAN OF THE DAY

Friday, July 09, 2010

OPERATIONS: Exploration Indonesia

	0000
	Underway as before
Conducting mapping operations

	0630
	Secure all overboard discharges

	0730
	Bow and stern thrusters online

	0745
	Safety Brief (Bridge)

	0815
	Secure mapping operations

Commence ROV deployment

	1500
	Ops Brief (Forward Lounge)

	1700
	ROV on deck
Resume mapping operations


· Continue ROV dive in DP
· Continue mapping operations
· 0030 conduct XBT
On Deck for Saturday:

· Conduct ROV operations

· Conduct multibeam and XBT operations overnight
==============================================================================
=========================END OF SITUATION REPORT==============================
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Stalked Sponge from the July 5 dive





Crinoid on sponge from the July 6 dive





Octopus from the July 8 dive








